Author's response to reviews

Title: Socioeconomic position and work, travel, and recreation-related physical activity in Japanese adults: a cross-sectional study

Authors:

Munehiro Matsushita (m.munehiro1987@akane.waseda.jp)
Kazuhiro Harada (haradak@ncgg.go.jp)
Takashi Arao (tarao@waseda.jp)

Version: 5
Date: 3 August 2015

Author's response to reviews: see over
Responses to the comments of Reviewer #3

1. In the abstract, please correct the last sentence: women working full-time have significantly lower and not higher travel related physical activity.

*Response:*
Thank you for your comment. We corrected the sentence.

(Abstract: Result)
There was no relation between working full time and physical activity in men, but women working full-time have significantly lower and not higher travel related physical activity. (OR 0.75; 0.59–0.96).

2. In the abstract it says that there were 3264 participants but in the methods it has been said that there were 3269 participants. Please revise this discrepancy.

*Response:*
Thank you for your comment. We corrected the sentence and re-calculated the mean and standard deviation of age.

(Abstract: Methods)
A total of 3269 subjects, 1651 men (mean ± standard deviation: 44.2 ± 8.1 years) and 1618 women (44.1 ± 8.1 years), responded to an Internet-based cross-sectional survey.

3. Line 73 : please correct > to <3million

*Response:*
Thank you for your comment. We corrected the sentence.

(p.6, Lines.73)
The socioeconomic variables included household income (categorized as < 3 million yen, 3–7 million yen, ≥ 7 million yen), educational level (junior high and high school graduation, 2-year college degree or equivalent, and 4-year college or higher degree),
and employment status (non full time worker or full time worker).